Author's response to reviews

Title: Femoral vein obturator bypass revascularization in septic groin infections: report of two cases

Authors:

Albert Busch (busch_a2@chirurgie.uni-wuerzburg.de)
Udo Lorenz (lorenz_u1@chirurgie.uni-wuerzburg.de)
Christian Tierbe (tiurbe_c@chirurgie.uni-wuerzburg.de)
Christoph Bühler (buehler_c@chirurgie.uni-wuerzburg.de)
Richard Kellersmann (kellersman_r@chirurgie.uni-wuerzburg.de)

Version: 2 Date: 1 July 2012

Author's response to reviews:

With this manuscript we hope to provide useful insight into the management of a growing number of patients presenting with life-threatening infectious arterial bleeding due to severe groin infections with highly pathogenic bacteria. The obturator bypass from autologous femoral vein is a definitive solution with the best outcome.

In this new version of the manuscript I rearranged the distinct sections' order refering to the comments from the initial review process and provided the additional information concerning, patients' data, authors' contributions, declaration of interests and informed consent.

I also revised the manuscripts title and abstract fitting more closely to BioMed Central's instructions for authors.

Thank you very much for providing useful advice and help